Author's response to reviews

Title: CT Colonography Imaging of Pneumatosis Intestinalis after Hyperbaric Oxygenotherapy

Authors:

jean louis frossard (jean-louis.frossard@hcuge.ch)
philippe braude (philippe.braude@colline.ch)
marc nicollazi (marc.nicallazi@gmail.com)
jean yves berney (jean-yves-berney@hcuge.ch)

Version: 2 Date: 7 February 2011

Author's response to reviews: see over
Hôpital Cantonal Universitaire de Genève
Division de Gastroentérologie

To The Editors

Genève, le 7 February 2011

N/Réf.: 

RE : submission of a revised version of our manuscript

Dear Editor,

Enclosed you will find our revised manuscript entitled “CT Colonography Imaging of Pneumatosis Intestinalis after Hyperbaric Oxygenotherapy” that we would like to be considered for publication in your Journal. Hereafter you will find our replies to the reviewers’s queries.

Point by point replies to the reviewers :

Reviewer E Schropfer:

As suggested the title of the paper has been changed to “CT Colonography Imaging of Pneumatosis Intestinalis after Hyperbaric Oxygenotherapy”. We also removed the words that were judged inappropriate as suggested.

We have also taken into account the suggestion for the discussion section that has been changed and we are now discussing some aspect of the treatment. We have to say that no controlled study does exist in the topic so the literature is weak.

Figure 1 has been reduced to panels ADE as recommended.
Reviewer M Hershman

We thank the reviewer for his nice comments. We are now quoting the suggested reference.

In the hope you will be interested in the content of our revised manuscript, I thank you in advance for your consideration

Sincerely yours

Jean Louis Frossard, MD